food & nutrition (

REVIEW ARTICLE Dietary patterns – A scoping review for Nordic Nutrition Recommendations 2023

Henna Vepsäläinen¹* and Jaana Lindström²

¹Department of Food and Nutrition, University of Helsinki, Helsinki, Finland; ²Finnish Institute for Health and Welfare, Helsinki, Finland

Popular scientific summary

- Dietary pattern refers to the quantities, proportions, variety, or combination of foods and drinks typically consumed.
- A dietary pattern can be identified according to predefined indices, data-driven methods, or the distribution of nutrients.
- Dietary patterns in the Nordic and Baltic countries have been characterised as Western/sweet.
- There is moderate to strong evidence that dietary patterns high in vegetables, fruits, whole grains, nuts, legumes, low-fat dairy and seafood, and low in red and processed meat, refined grains, and sugar-sweetened foods and beverages are associated with decreased risk of all-cause mortality, CVD, obesity, type 2 diabetes, and breast and colorectal cancer.

Abstract

Background: A dietary pattern can be defined as the quantities, proportions, variety, or combination of foods and drinks typically consumed. The dietary pattern approach aims to place emphasis on the total diet as a long-term health determinant, instead of focussing on separate foods and nutrients, which may interact or confound each other.

Aim: This scoping review describes the totality of evidence for the role of dietary patterns for health-related outcomes as a basis for setting and updating food-based dietary guidelines in the Nordic Nutrition Recommendations 2023 (NNR2023).

Methods: We used evidence from 10 qualified systematic reviews identified by the NNR2023 project. No additional literature search was conducted.

Results: Strong or moderate evidence linked dietary patterns high in vegetables, fruits, whole grains, fish, low-fat dairy and legumes, and low in red and processed meats, sugar-sweetened beverages, sugary foods- and refined grains with beneficial health outcomes, such as reduced risk of cardiovascular disease (CVD), type 2 diabetes, obesity, cancer, bone health, and premature death. We also found limited evidence suggesting a relationship with the described dietary patterns in childhood and decreased risk of obesity and hypertension later in life. Most studies have been conducted among adult populations, and thus, there is a need for studies in certain subgroups, such as children and adolescents as well as the elderly.

Keywords: whole diet; eating habits; food consumption; healthy diet

Received: 17 May 2022; Revised: 26 September 2022; Accepted: 24 January 2024; Published: 30 April 2024

Traditionally, research on the health effects of diet has focused on the relationships between nutrients, foods, or food groups, and health outcomes. However, as people eat foods and beverages in combinations, epidemiological studies have shifted their attention towards dietary patterns (1). A dietary pattern can be defined as the quantities, proportions, variety, or combination of foods and drinks a person typically consumes. Dietary patterns can be identified using predefined indices, which describe how closely a person's diet resembles the given diet (2). For example, the Healthy Eating Index (HEI), the Mediterranean Diet Score (MedDiet Score), and the Healthy Nordic Diet have been developed to describe healthfulness of diets in different populations.

Box 1. Background papers for Nordic Nutrition Recommendations 2023

- This paper is one of many scoping reviews commissioned as part of the Nordic Nutrition Recommendations 2023 (NNR2023) project (7).
- The papers are included in the extended NNR2023 report but, for transparency, these scoping reviews are also published in Food & Nutrition Research.
- The scoping reviews have been peer reviewed by independent experts in the research field according to the standard procedures of the journal.
- The scoping reviews have also been subjected to public consultations (see report to be published by the NNR2023 project).
- The NNR2023 committee has served as the editorial board.
- While these papers are a main fundament, the NNR2023 committee has the sole responsibility for setting dietary reference values in the NNR2023 project.

The indices typically share similar characteristics, such as high consumption of fruits and vegetables, whole grains, fish, and unsaturated fats, and low consumption of sweets and sugar-sweetened beverages, but differ on some details because of cultural and food environmental differences between populations. For instance, the Mediterranean diet emphasises the use of olive oil (3), whereas the Healthy Nordic Diet recommends the use of rapeseed oil (4). The adherence to these dietary patterns is typically measured with a score, which can be interpreted to describe the healthfulness of the diet.

Another approach is to use data-driven methods, such as principal component analysis (PCA), factor analysis, or cluster analysis (5). These methods do not depend on the researcher's definition of a healthful diet but are instead based on intercorrelations between food items (PCA, factor analysis) or individual differences in mean consumption (cluster analysis) (6). The data-driven methods describe the actual food consumption patterns in the given sample, and the patterns derived may be more or less healthy or unhealthy. A third approach to dietary patterns is to address the nutrient distribution of habitual diet, where distinction is made when, for example, at least one of the macronutrients (carbohydrates, fat, or protein) is outside the "acceptable macronutrient distribution range." In the present paper, dietary patterns based solely on micronutrient intakes are not discussed, because the same micronutrients can be obtained from very different food sources, and thus, micronutrient intake distributions are not representative of food-level dietary patterns.

Dietary pattern analysis approach is advantageous since nutrients available in food may confound or interact with each other (1). Additionally, pattern analysis can be considered useful in evaluating dietary guidelines and may enable the detection of associations between diet and health outcomes, as the combined effect of an entire diet may be more powerful than the effects of its individual components (1, 2). Moreover, the whole-diet approach may be easier for the consumers to comprehend and implement in their everyday life. However, the dietary pattern approach also has its limitations. Most importantly, the identification of data-driven dietary patterns is strongly dependent on subjective and arbitrary decisions about combining of food groups and the labelling of dietary patterns (1). Furthermore, associated health outcomes may also be driven by single nutrients or foods included in the dietary pattern and not by the patterns in general, and thus, dietary pattern analysis should be considered as a complementary approach facilitating the understanding of diet-health-relationships (1). This scoping review summarises the evidence on the link between dietary patterns or whole-diet and health-related outcomes as a basis for setting and updating food-based dietary guidelines (FBDGs) in the Nordic Nutrition Recommendations (NNR) 2023 (Box 1). Both theory-based and data-driven dietary patterns were included.

Methods

The scoping review follows closely the protocol developed within the NNR2023 project (7). The sources of evidence used in the review follow the eligibility criteria described in an earlier paper (8). No *de novo* NNR2023 systematic reviews were conducted for the purposes of this paper, but altogether 10 qualified systematic reviews were recognised as suitable evidence (9). These qualified systematic reviews were used as primary evidence in this review.

Diet intake in Nordic and Baltic countries

Most of the Nordic and Baltic countries monitor dietary intake among adult population, but dietary patterns are not routinely tracked, and the countries lack comprehensive, structured information on dietary patterns. Unlike predefined dietary indices, such as the Healthy Nordic Diet, data-driven dietary patterns, which describe the real food consumption patterns in a given population, can be used to describe dietary patterns on a general level. A few studies, examples of which are briefly described in the following paragraphs, have provided information about the dietary patterns in the Nordic and Baltic countries using predominantly PCA or factor analysis.

In general, it seems that "healthy" or "prudent" patterns characterised by frequent intake of for example vegetables, fruits, high-fibre bread, and vegetable oils have been identified in Swedish (10–14), Norwegian (13, 15, 16), Danish (13, 17), and Finnish (14, 18) adult populations. In addition, a "traditional" or "mixed" dietary pattern characterised by frequent consumption of for example potatoes, meat, bread, and fish has shown to be evident in Sweden (10, 11, 14), Norway (13, 15), Denmark (13), and Finland (14, 18). The "traditional" and "mixed" dietary patterns usually share traits of both healthy and unhealthy diets, and thus, they cannot be easily categorised as healthy or unhealthy.

A frequently identified dietary pattern in the Nordic countries is also a "Western" or a "sweet" dietary pattern, which consists of foods typically considered unhealthy, such as fried potatoes and fast food, meat products, sweets, and soft drinks (11-13, 15, 16, 18). Interestingly, some studies have also identified an "alcohol" pattern, which seems to be characterised by frequent consumption of, for example, meat products and salty snacks, on top of alcohol usage (14, 16, 17). In a few studies, alcohol consumption has been associated with an unhealthy dietary pattern (12, 13, 18), but it has also been shown to relate to a traditional pattern among Norwegian Sami population (15) and to a prudent pattern among Finnish middle-aged men (18). When analysed separately, similar patterns seem to emerge both among men and women (e.g. "healthy" or "traditional") (10, 12), but in general women tend to score higher or be over-represented in the "healthy" patterns (11, 15, 17).

Studies among children and adolescents have yielded rather similar patterns apart from the alcohol-related dietary patterns. A "healthy" pattern seems to be most prominent and has been identified in Finnish (19, 20), Icelandic (21), Estonian, and Swedish (22) populations. In addition, "unhealthy" (19, 20, 22) and "traditional" or "mixed" (19–21) patterns have also been found among children in the Nordic and Baltic countries.

Health outcomes relevant for Nordic and Baltic countries

Obesity

In adults, moderate evidence from 54 longitudinal studies showed an association between favourable outcomes related to body weight (e.g. lower body mass index [BMI], waist circumference) or risk of obesity and dietary patterns characterised by higher intakes of vegetables, fruits, whole grains, seafood, and legumes; moderate intakes of dairy products (with emphasis on low- and non-fat dairy) and alcohol; lower intakes of meats (including red and foods and beverages and refined grains (23). The foods included in the beneficial patterns contain higher levels of unsaturated fats and lower levels of saturated fats, cholesterol, and sodium. A systematic review of 12 longitudinal studies concluded that there is limited evidence linking dietary patterns lower in fruits, vegetables, whole grains, and low-fat dairy and higher in added sugars, refined grains, fried potatoes, and processed meats with higher BMI in adolescence (23). The 2020 Dietary Guidelines Advisory Committee (USA) could not evaluate the possible associations between diets based on macronutrient distribution and obesity because of insufficient evidence.

processed meats); and low intakes of sugar-sweetened

Cardiovascular diseases

Altogether 149 articles examining dietary patterns and risk of CVDs among adults were included in a systematic review published by the 2020 Dietary Guidelines Advisory Committee (24). The authors concluded that strong and consistent evidence links decreased risk of CVD with dietary patterns characterised by higher consumption of vegetables, fruits, whole grains, low-fat dairy, and seafood and lower consumption of red and processed meat, refined grains, and sugar-sweetened foods and beverages. Most studies have also shown regular consumption of nuts and legumes and moderate consumption of alcohol to be beneficial in reducing CVD risk. A systematic review of four longitudinal studies provided limited evidence showing that lower blood pressure and blood lipid levels later in life associated with dietary patterns higher in vegetables, fruits, whole grains, fish, low-fat dairy and legumes, and lower in sugar-sweetened beverages, other sweets, and processed meat among children and adolescents (24). Non-energy restricted diets based on macronutrient distribution with either carbohydrate, fat, and/or protein proportion outside of the "acceptable macronutrient distribution range" were judged to be neither beneficial nor detrimental among adults, albeit the evidence was limited because of, for example, limitations in the study designs. No studies investigating CVDs and diets based on macronutrient distribution consumed by children were identified.

Type 2 diabetes

The systematic review (25) identified only one article exploring (retrospectively) the association between dietary patterns in adolescence and risk of type 2 diabetes by middle age. The evidence to give recommendations was considered inadequate. As regards to adults, 52 articles were identified in the literature search covering years 2014 to 2020. The authors concluded that there is moderate evidence to indicate that dietary patterns higher in vegetables, fruits and whole grains, and lower in red and processed meats, high-fat dairy products, refined grains

and sweets/sugar-sweetened beverages reduce the risk of developing type 2 diabetes. The authors identified 23 articles (2 randomised clinical trials (RCTs), 21 prospective cohort studies) published between 2000 and 2020 where diets with differing macronutrient distributions were studied. Most of the studies explored diet with proportion of carbohydrates below and fat above the acceptable range. The authors concluded that it is not possible to make recommendations about macronutrient distribution and risk of type 2 diabetes based on the available evidence.

Cancer

In a systematic review of 26 studies (3 RCTs, 21 prospective cohorts, and 2 nested case-control studies), dietary patterns characterised by high intakes of vegetables, fruits and whole grains, and low intakes of animal-source foods and refined carbohydrates were linked to reduced risk of postmenopausal breast cancer (26). The evidence was graded moderate. The evidence regarding dietary patterns and premenopausal breast cancer risk was limited but parallel.

By systematically reviewing 24 studies (2 RCTs, 21 prospective cohorts, and 1 nested case-control study), the 2020 Dietary Guidelines Advisory Committee concluded that moderate evidence links dietary patterns higher in vegetables, fruits, legumes, whole grains, lean meats and seafood, and low-fat dairy, and low in red and processed meats, saturated fat, sugar-sweetened beverages, and sweets with lower risk of colon and rectal cancer (26). Moderate evidence showed that dietary patterns higher in red and processed meats, French fries, potatoes, and sources of sugars (e.g. sugar-sweetened beverages, sweets and desserts) associated with a greater colon and rectal cancer risk.

Regarding lung cancer, the 2020 Dietary Guidelines Advisory Committee reviewed seven prospective cohort studies and one nested case-control study (26). The authors concluded that limited evidence shows a relationship between dietary patterns higher in vegetables, fruits, seafood, grains and cereals, legumes and lean meats, and lower in fat or non-fat dairy products with lower risk of lung cancer. The association was evident primarily among former and current smokers. A systematic review of seven prospective cohort studies and one nested case-control study yielded limited evidence suggesting no relationship between dietary patterns and risk of prostate cancer (26).

Bone health

In a systematic review of nine prospective cohort studies, dietary patterns characterised by high consumption of fruits, vegetables, legumes, nuts, low-fat dairy, whole grains and fish, and lower consumption of meats (particularly processed meats), sugar-sweetened beverages, and sweets was linked to beneficial bone health outcomes in adults (27). The main bone health outcome used in the studies was hip fractures, and the evidence was graded moderate. Only two articles examining the link between dietary patterns and bone health outcomes in adolescents were identified. Thus, the relationship between dietary patterns in adolescence and later bone health could not be estimated.

Neurocognitive health

The 2020 Dietary Guidelines Advisory Committee conducted a systematic literature search to identify studies evaluating the association between dietary patterns and neurocognitive health (28). Altogether 26 studies (4 RCTs, 22 observational studies) using cognitive performance, age-related cognitive decline, mild cognitive impairment and/or incident dementia as outcome and published between January 2014 and February 2020 were identified. In the studies, both exposures and outcomes were measured using various approaches. The conclusion was that there is limited evidence suggesting that dietary patterns containing vegetables, fruits, unsaturated vegetable oils and/or nuts, legumes, and fish or seafood consumed during adulthood are associated with lower risk of age-related cognitive impairment and/or dementia.

Other health outcomes

The 2020 Dietary Guidelines Advisory Committee systematically reviewed four prospective cohort studies examining the association between dietary patterns and sarcopenia (29). However, because of methodological inconsistencies and risk of bias, the authors concluded the evidence to be insufficient to determine the relationship between dietary patterns and sarcopenia.

A total of 26 articles (mostly from prospective cohort studies) were included in a systematic review investigating the association between dietary patterns during pregnancy and risk of excessive gestational weight gain during pregnancy (30). The review concluded that limited evidence links dietary patterns higher in vegetables, fruits, nuts, legumes and fish, and lower in added sugar, and red and processed meat with lower risk of excessive gestational weight gain during pregnancy. Strong conclusions could not be drawn because of lack of RCTs and controlling for key confounders, risk of bias, and unrepresentativeness of the samples.

In a systematic review of three RCTs and two cross-sectional studies, the 2020 Dietary Guidelines Advisory Committee concluded that limited evidence supports the finding that maternal dietary patterns associate with the relative proportions of saturated fat, monounsaturated fatty acids, and polyunsaturated fatty acids in human milk (31). However, because of the small number of studies included and methodological differences in defining dietary patterns, the characteristics of beneficial dietary patterns could not be unambiguously described. In addition, the included studies had small and unrepresentative samples, did not control for key confounders, and were heterogeneous in terms of timing and methods of human

Mortality

The 2020 Dietary Guidelines Advisory Committee reported in a systematic review consisting mostly of prospective cohort studies (140 studies out of the 141 included studies) that dietary patterns characterised by frequent consumption of vegetables, fruits, legumes, nuts, whole grains, unsaturated vegetable oils, and fish, lean meat, or poultry, when meat was included, associated with decreased risk of all-cause mortality (32). The patterns contained relatively low amounts of red and processed meat, high-fat dairy, and refined carbohydrates or sweets. The evidence was ranked strong and demonstrated within adult and older adult populations. The link between diets based on macronutrient distribution and all-cause mortality could not be assessed because of insufficient evidence.

milk collection. With regards to other milk composition

parameters and quantity, no evidence was available.

Mechanisms

The recommended dietary patterns are typically nutrient-dense and include a variety of food groups that have been found to associate with health benefits. For example, consumption of fruit, vegetables, and wholegrain cereals have been linked to reduced risk of CVD, cancer, and all-cause mortality (33, 34). Both fruits and vegetables as well as wholegrain products contain a wide range of nutrients and phytochemicals, such as fibre, vitamin C, carotenoids, potassium, antioxidants, flavonoids, and other compounds, which can act to reduce the risk of chronic diseases through several biological mechanisms. These mechanisms can include, for example, increased satiety (35), modulation of stress- and defence-related gene expression (36), promotion of beneficial microbial metabolites (37), or reduction of low-grade inflammation (38). Additionally, frequent fish consumption has also been linked to health benefits (39). Fatty fish is a significant source of omega-3 fatty acids, especially eicosapentaenoic acid (EPA) and docosahexaenoic acid (DHA), which have been shown to, for example, prevent inflammation and decrease triglyceride concentration through several mechanisms including alterations in cell membrane composition and eicosanoid production (40).

The recommended dietary patterns usually include restricted amounts of processed or red meat and sugar-sweetened beverages. High consumptions of processed and red meat have been linked to increased risk of cancer, type 2 diabetes, and mortality (41). The mechanism linking processed meat consumption with health outcomes remains unclear, but for example high levels of sodium (42), polycyclic aromatic hydrocarbons, and heterocyclic aromatic amines as well as nitrosyl-heme have been hypothesised to play a role (43). High intake of sugar-sweetened beverages has been shown to associate with weight gain and obesity (44) through increased energy intake (45). Sugar-sweetened beverage consumption may also link to other adverse health outcomes, such as greater incidence of type 2 diabetes (46). High consumption of sugar-sweetened beverages can also associate with decreased bone health through multiple potential mechanisms including sugar, phosphate, acidity, and caffeine, which can all negatively affect bone metabolism (47). Dietary patterns that include large amounts of energy-dense foods with low nutrient density (e.g. sugar-sweetened beverages and other foods high in sugar, sodium, and/or saturated fat) may also lead to micronutrient deficiencies, which in turn can predispose to chronic diseases.

To sum up, multiple mechanisms mediating the beneficial and harmful effects of food groups, foods, and nutrients have been identified and probably contribute to the health effects of dietary patterns. Moreover, nutrients are ingested as a part of the food matrix and composite meals, which may modulate their absorption and functions in the body. Nutrients and other compounds in the diet may have synergistic, antagonistic, or other interactive effects as part of the whole diet. Some of these effects remain unknown, and thus, they cannot be measured or controlled for.

Food-based dietary guidelines

In general, dietary patterns higher in vegetables, fruits, whole grains, fish, low-fat dairy and legumes, and lower in red and processed meats, sugar-sweetened beverages, sugary foods (e.g. sweets), and refined grains were found to be protective in terms of CVD, type 2 diabetes, obesity, breast and colon cancer, bone health, and premature death among adult populations. The evidence was graded as strong or moderate. Limited evidence was found for the protective effect of the aforementioned dietary patterns regarding lung and prostate cancer, neurocognitive health, and gestational weight gain. Similarly, limited evidence linked the described dietary patterns in childhood and/or adolescence with smaller risk of obesity and hypertension later in life. It was not possible to make recommendations about macronutrient distribution and risk of chronic diseases based on the available evidence.

The beneficial dietary patterns are characterised by food items that have been linked to improved health outcomes, such as vegetables, fruits, whole grains, and legumes. These foods are in general high in micronutrients and contain low or moderate amounts of energy. Similarly, the patterns typically contain only small amounts of food items generally linked with unbeneficial health outcomes, such as red and processed meat and sugar-sweetened beverages. Red and processed meat may link to poor health via various pathways, whereas sugar-sweetened beverages contain virtually no micronutrients but may contribute to excess energy intake. Excess energy intake will, in turn, lead to obesity, which is an important risk factor for cardiometabolic diseases and several types of cancer (48).

Dietary fat quality can contribute to the risk of several chronic diseases, as mentioned earlier. In general, the foods included in the beneficial patterns contain higher levels of unsaturated and lower levels of saturated fats. Specifically fish, which is a good source of unsaturated fatty acids, was included in the healthy dietary pattern in many of the qualified systematic reviews. Inclusion of vegetable oils with high unsaturated and low saturated fat content in recommended dietary patterns could further improve dietary fat quality.

Dietary patterns higher in plant-based foods and lower in animal-based foods are environmentally more sustainable than patterns high in animal-based foods (49). As the pattern identified in this scoping review emphasises plant-based foods (vegetables, fruits, whole grains), it can be argued that compared to current diets in the Nordic and Baltic countries, the recommended pattern would be beneficial for public health and environmental sustainability. However, the recommended pattern also includes animal-based foods, such as fish and low-fat dairy products. It is worth noticing that dietary patterns are very broad descriptions of diet and cannot be interpreted as exact recommendations on the composition of a healthy diet. These findings do not indicate that a dietary pattern must contain animal-based foods to be healthy, nor do they show that low levels of meat consumption would be disadvantageous.

The methods used to derive dietary patterns are mainly designed to capture the existing diets of the populations, which typically contain both plant- and animal-based foods. Thus, based on the evidence presented by the qualified systematic reviews, conclusions on the healthiness of vegan or vegetarian diets as opposed to mainly plantbased diets that may also contain fish, low-fat dairy products, and low amounts of meat and/or other animal-based foods (the recommended diet) cannot be drawn. As for example in Finland, the prevalence of vegetarianism has increased from 0.7 to 1.8% between 2012 and 2017 (50) and may continue to rise, future studies should aim to examine dietary patterns among vegans and vegetarians and compare their health effects with those of a healthy, mixed dietary pattern.

Many of the qualified systematic reviews used as sources of evidence in this scoping review were able to identify studies in both adult and child populations. However, the evidence regarding the associations between childhood dietary patterns and later health outcomes were mostly limited because of the small number of prospective studies. Thus, more longitudinal studies tracking dietary patterns and health outcomes from childhood are needed. Moreover, studies investigating osteoporosis and sarcopenia were very few, and more studies are also needed among the elderly. It is noteworthy that the majority of the studies included in the qualified systematic reviews were observational, which limits the conclusions. However, as the considered health outcomes take multiple years to develop, well-designed, good-quality prospective studies are in many cases the best possible evidence.

The scoping review used 10 qualified systematic reviews identified by an initial literature search as sources of evidence. The qualified systematic reviews were conducted by the United States Department of Agriculture (USDA) and were of good quality. However, most of them covered the timeframe up until 2019. Thus, studies published in 2019–2022 have not been covered. It is unlikely that during that time, such a large body of conflicting evidence would have accumulated that the strong or moderate evidence had to be downgraded into limited. By contrast, it is possible that some of the limited evidence would now be graded as moderate.

As mentioned in the introduction, because of the complex associations between nutrients, foods, diets, and health outcomes, the dietary pattern analysis should be considered as a complementary approach when examining the relations between diet and health (1). Furthermore, as dietary assessment methods and food groups included in the analysis influence the results, comparison of dietary patterns between study populations is challenging. Thus, the method of constructing the dietary patterns may lead to unspecific and sometimes even partly incorrect conclusions. To create more specific knowledge on dietary patterns, future studies should consider focusing on more detailed food groups, such as subtypes of vegetables, fried and boiled/mashed potatoes, red meat and processed meat, or different types of dairy products. However, as dietary patterns reflect the whole diet of the participants rather than the intake of specific foods or nutrients, the level of detail in analysis cannot be similar to studies investigating the associations between single nutrients or foods and health outcomes.

Conflict of interest and funding

The authors have not received any funding or benefits from industry to conduct this study, and report no conflicts of interests. The authors received a small reimbursement from the Norwegian Directorate of Health for work linked to this article.

References

 Mozaffarian D, Rosenberg I, Uauy R. History of modern nutrition science – implications for current research, dietary guidelines, and food policy. BMJ 2018; 361: k2392. doi: 10.1136/bmj. k2392

- Willett W. Issues in analysis and presentation of dietary data. In: Willett W, ed. Nutritional epidemiology. New York: Oxford University Press; 2012, pp. 305–33.
- Davis C, Bryan J, Hodgson J, Murphy K. Definition of the Mediterranean Diet; a literature review. Nutrients 2015; 7: 9139–53. doi: 10.3390/nu7115459
- Uusitupa M, Hermansen K, Savolainen MJ, Schwab U, Kolehmainen M, Brader L, et al. Effects of an isocaloric healthy Nordic diet on insulin sensitivity, lipid profile and inflammation markers in metabolic syndrome – a randomized study (SYSDIET). J Intern Med 2013; 274(1): 52–66. doi: 10.1111/joim.12044
- Hu B. Dietary pattern analysis: a new direction in nutritional epidemiology. Curr Opin Lipidol 2002; 13: 3–9. doi: 10.1097/00041433-200202000-00002
- Newby PK, Tucker KL. Empirically derived eating patterns using factor or cluster analysis: a review. Nutr Rev 2004; 62: 177–203. doi: 10.1111/j.1753-4887.2004.tb00040.x
- Blomhoff R, Andersen R, Arnesen EK, Christensen JJ, Eneroth H, Erkkola M, et al. Nordic Nutrition Recommendations 2023. Copenhagen: Nordic Council of Ministers; 2023.
- Christensen JJ, Arnesen EK, Andersen R, Enerot H, Erkkola M, Høyer A, et al. The Nordic Nutrition Recommendations 2022 – principles and methodologies. Food Nutr Res 2020; 64. doi: 10.29219/fnr.v64.4402
- Høyer A, Christensen JJ, Arnesen EK. The Nordic Nutrition Recommendations 2022 – Prioritisation of topics for de novo systematic reviews. Food Nutr Res 2021; 65: 7828. doi: 10.29219/fnr.v65.7828
- Ax E, Warensjö Lemming E, Becker W, Anderson A, Lindroos AK, Cederholm T, et al. Dietary patterns in Swedish adults; results from a national dietary survey. Br J Nutr 2016; 115: 95–104. doi: 10.1017/S0007114515004110
- Berg CM, Lappas G, Strandhagen E, Wolk A, Torén K, Rosengren A, et al. Food patterns and cardiovascular disease risk factors: the Swedish INTERGENE research program. Am J Clin Nutr 2008; 88: 289–97. doi: 10.1093/ajcn/88.2.289
- Huseinovic E, Hörnell A, Johansson I, Esberg A, Lindahl B, Winkvist A. Changes in food intake patterns during 2000–2007 and 2008–2016 in the population-based Northern Sweden Diet Database. Nutr 2019; 18: 36. doi: 10.1186/s12937-019-0464-0
- Engeset D, Hofoss D, Nilsson LM, Olsen A, Tjønneland A, Skeie G. Dietary patterns and whole grain cereals in the Scandinavian countries – differences and similarities. The HELGA project. Public Health Nutr 2014; 18: 905–15. doi: 10.1017/S1368980014001104
- Balder HF, Virtanen M, Brants HAM, Krogh V, Dixon LB, Tan F, et al. Common and country-specific dietary patterns in four European cohort studies. J Nutr 2003; 133: 4246–51. doi: 10.1093/jn/133.12.4246
- Petrenya N, Rylander C, Brustad M. Dietary patterns of adults and their associations with Sami ethnicity, sociodemographic factors, and lifestyle factors in a rural multiethnic population of northern Norway – the SAMINOR 2 clinical survey. BMC Public Health 2019; 19: 1632. doi: 10.1186/s12889-019-7776-z
- Engeset D, Alsaker E, Ciampi A, Lund E. Dietary patterns and lifestyle factors in the Norwegian EPIC cohort: the Norwegian Women and Cancer (NOWAC) study. Eur J Clin Nutr 2005; 59: 675–84. doi: 10.1038/sj.ejcn.1602129
- Frederiksen SB, Themsen HH, Overvad K, Dahm C. Dietary patterns generated by the Treelet Transform and risk of stroke: a Danish cohort study. Public Health Nutr 2020; 24: 84–94. doi: 10.1017/S1368980019004324

- Ruusunen A, Lehto SM, Mursu J, Tolmunen T, Tuomainen T, Kauhanen J, et al. Dietary patterns are associated with the prevalence of elevated depressive symptoms and the risk of getting a hospital discharge diagnosis of depression in middle-aged or older Finnish men. J Affect Disord 2014; 159: 1–6. doi: 10.1016/j.jad.2014.01.020
- Vepsäläinen H, Korkalo L, Mikkilä V, Lehto R, Ray C, Nissinen K, et al. Dietary patterns and their associations with home food availability among Finnish pre-school children: a cross-sectional study. Public Health Nutr 2018; 21: 1232–42. doi: 10.1017/S1368980017003871
- Wu F, Pahkala K, Juonala M, Rovio SP, Sabin MA, Rönnemaa T, et al. Dietary pattern trajectories from youth to adulthood and adult risk of impaired fasting glucose: a 31-year cohort study. J Clin Endocr Metab 2021; 106: e2078–86. doi: 10.1210/clinem/ dgab044
- Thordardottir GS, Einarsdottir K, Thordardottir M, Tryggvadottir L, Valdimarsdottir UA, Gudnason V, et al. Dietary patterns in adolescence and risk of colorectal cancer: a population-based study. Cancer Causes Control 2022; 33: 205–11. doi: 10.1007/s10552-021-01524-z
- 22. Fernández-Alvira JM, Bammann K, Pala V, Krogh V, Barba G, Eiben G, et al. Country-specific dietary patterns and associations with socioeconomic status in European children: the IDEFICS study. Eur J Clin Nutr 2014; 68: 811–21. doi: 10.1038/ejcn.2014.78
- 23. 2020 Dietary Guidelines Advisory Committee and Nutrition Evidence Systematic Review Team. Dietary patterns and growth, size, body composition, and/or risk of overweight or obesity: a systematic review. 2020 Dietary Guidelines Advisory Committee Project. Alexandria, VA: U.S. Department of Agriculture, Food and Nutrition Service, Center for Nutrition Policy and Promotion; 2020. Available from: https://nesr.usda. gov/2020-dietary-guidelines-advisory-committee-systematic-reviews [cited 13 May 2022].
- 24. 2020 Dietary Guidelines Advisory Committee and Nutrition Evidence Systematic Review Team. Dietary patterns and risk of cardiovascular disease: a systematic review. 2020 Dietary Guidelines Advisory Committee Project. Alexandria, VA: U.S. Department of Agriculture, Food and Nutrition Service, Center for Nutrition Policy and Promotion; 2020. Available from: https://nesr.usda.gov/2020-dietary-guidelines-advisory-committee-systematic-reviews [cited 13 May 2022].
- 25. 2020 Dietary Guidelines Advisory Committee and Nutrition Evidence Systematic Review Team. Dietary patterns and risk of type 2 diabetes: a systematic review. 2020 Dietary Guidelines Advisory Committee Project. Alexandria, VA: U.S. Department of Agriculture, Food and Nutrition Service, Center for Nutrition Policy and Promotion; 2020. Available from: https://nesr.usda.gov/2020-dietary-guidelines-advisory-committee-systematic-reviews [cited 13 May 2022].
- 26. 2020 Dietary Guidelines Advisory Committee and Nutrition Evidence Systematic Review Team. Dietary patterns and breast, colorectal, lung, and prostate cancer: a systematic review. 2020 Dietary Guidelines Advisory Committee Project. Alexandria, VA: U.S. Department of Agriculture, Food and Nutrition Service, Center for Nutrition Policy and Promotion; 2020. Available from: https://nesr.usda. gov/2020-dietary-guidelines-advisory-committee-systematic-reviews [cited 13 May 2022].
- 27. 2020 Dietary Guidelines Advisory Committee and Nutrition Evidence Systematic Review Team. Dietary patterns and bone health: a systematic review. 2020 Dietary Guidelines Advisory

Committee Project. Alexandria, VA: U.S. Department of Agriculture, Food and Nutrition Service, Center for Nutrition Policy and Promotion; 2020. Available from: https://nesr.usda.gov/2020-dietary-guidelines-advisory-committee-systemat-ic-reviews [cited 13 May 2022].

- 28. 2020 Dietary Guidelines Advisory Committee and Nutrition Evidence Systematic Review Team. Dietary patterns and neurocognitive health: a systematic review. 2020 Dietary Guidelines Advisory Committee Project. Alexandria, VA: U.S. Department of Agriculture, Food and Nutrition Service, Center for Nutrition Policy and Promotion, July. Available from: https://nesr.usda.gov/2020-dietary-guidelines-advisory-committee-systematic-reviews [cited 13 May 2022].
- 29. 2020 Dietary Guidelines Advisory Committee and Nutrition Evidence Systematic Review Team. Dietary patterns and sarcopenia: a systematic review. 2020 Dietary Guidelines Advisory Committee Project. Alexandria, VA: U.S. Department of Agriculture, Food and Nutrition Service, Center for Nutrition Policy and Promotion; 2020. Available from: https://nesr.usda. gov/2020-dietary-guidelines-advisory-committee-systematic-reviews [cited 13 May 2022].
- 30. 2020 Dietary Guidelines Advisory Committee and Nutrition Evidence Systematic Review Team. Dietary patterns during pregnancy and gestational weight gain: a systematic review. 2020 Dietary Guidelines Advisory Committee Project. Alexandria, VA: U.S. Department of Agriculture, Food and Nutrition Service, Center for Nutrition Policy and Promotion; 2020. Available from: https://nesr.usda.gov/2020-dietary-guidelines-advisory-committee-systematic-reviews [cited 13 May 2022].
- 31. 2020 Dietary Guidelines Advisory Committee and Nutrition Evidence Systematic Review Team. Dietary patterns during lactation and human milk composition and quantity: a systematic review. 2020 Dietary Guidelines Advisory Committee Project. Alexandria, VA: U.S. Department of Agriculture, Food and Nutrition Service, Center for Nutrition Policy and Promotion; 2020. Available from: https://nesr.usda.gov/2020-dietary-guidelines-advisory-committee-systematic-reviews [cited 13 May 2022].
- 32. 2020 Dietary Guidelines Advisory Committee and Nutrition Evidence Systematic Review Team. Dietary patterns and all-cause mortality: a systematic review. 2020 Dietary Guidelines Advisory Committee Project. Alexandria, VA: U.S. Department of Agriculture, Food and Nutrition Service, Center for Nutrition Policy and Promotion; 2020. Available from: https://nesr.usda.gov/2020-dietary-guidelines-advisory-committee-systematic-reviews [cited 13 May 2022].
- 33. Aune D, Giovannucci E, Boffetta P, Fadnes LT, Keum N, Norat T, et al. Fruit and vegetable intake and the risk of cardiovascular disease, total cancer and all-cause mortality a systematic review and dose-response meta-analysis of prospective studies. Int J Epidemiol 2017; 46: 1029–56. doi: 10.1093/ije/dyw319
- Chen G, Tong X, Xu J, Han S, Wan Z, Qin J, et al. Whole-grain intake and total, cardiovascular, and cancer mortality: a systematic review and meta-analysis of prospective studies. Am J Clin Nutr 2016; 104: 164–72. doi: 10.3945/ajcn.115.122432
- Warrilow A, Mellor D, McKune A, Pumpa K. Dietary fat, fibre, satiation, and satiety – a systematic review of acute studies. Eur J Clin Nutr 2019; 73: 333–44. doi: 10.1038/s41430-018-0295-7
- 36. Bøhn SK, Myhrstad MC, Thoresen M, Holden M, Karlsen A, Haugen Tunheim S, et al. Blood cell gene expression associated with cellular stress defense is modulated by antioxidant-rich food in a randomised controlled clinical trial of male smokers. BMC Med 2010; 8: 54. doi: 10.1186/1741-7015-8-54

- Shanahan F, van Sinderen D, O'Toole PW, Stanton C. Feeding the microbiota: transducer of nutrient signals for the host. Gut 2017; 66: 1709–17. doi: 10.1136/gutjnl-2017-313872
- Roager HM, Vogt JK, Kristensen M, Hansen LBS, Ibrügger S, Mærkedahl RB, et al. Whole grain-rich diet reduces body weight and systemic low-grade inflammation without inducing major changes of the gut microbiome: a randomised cross-over trial. Gut 2019; 68: 83–93. doi: 10.1136/gutjnl-2017-314786
- Krittanawong C, Isath A, Hahn J, Wang Z, Narasimhan B, Kaplin SL, et al. Fish consumption and cardiovascular health: a systematic review. Am J Med 2021; 134: 713–20. doi: 10.1016/j. amjmed.2020.12.017
- Riediger ND, Othman RA, Suh M, Moghadasian MH. A systematic review of the roles of n-3 fatty acids in health and disease. J Am Diet Assoc 2009; 109: 668–79. doi: 10.1016/j.jada.2008.12.022
- Yip CSC, Lam W, Fielding R. A summary of meat intakes and health burdens. Eur J Clin Nutr 2018; 72: 18–29. doi: 10.1038/ ejcn.2017.117
- 42. Kim Y, Keogh J, Clifton P. A review of potential metabolic etiologies of the observed association between red meat consumption and development of type 2 diabetes mellitus. Metabolism 2015; 64: 768–79. doi: 10.1016/j.metabol.2015.03.008
- Demeyer D, Mertens B, De Smet S, Ulens M. Mechanisms linking colorectal cancer to the consumption of (processed) red meat: a review. Crit Rev Food Sci Nutr 2016; 56: 2747–66. doi: 10.1080/10408398.2013.873886
- 44. Malik VS, Schulze MB, Hu FB. Intake of sugar-sweetened beverages and weight gain: a systematic review. Am J Clin Nutr 2006; 84: 274–88. doi: 10.1093/ajcn/84.2.274
- 45. Schulze MB, Manson JE, Ludwig DS, Colditz GA, Stampfer MJ, Willett WC, et al. Sugar-sweetened beverages, weight gain, and incidence of type 2 diabetesw in young and middle-aged women. JAMA 2004; 292: 927–34. doi: 10.1001/jama.292.8.927
- 46. Imamura F, O'Connor L, Ye Z, Mursu J, Hayashino Y, Bhypathiraju SN, et al. Consumption of sugar sweetened beverages, artificially sweetened beverages, and fruit juice and incidence of type 2 diabetes: systematic review, meta-analysis, and estimation of population attributable fraction. BMJ 2015; 351: h3576. doi: 10.1136/bmj.h3576
- Ahn H, Kyoung Park Y. Sugar-sweetened beverage consumption and bone health: a systematic review and meta-analysis. Nutr J 2021; 20: 41. doi: 10.1186/s12937-021-00698-1
- 48. Guh DP, Zhang W, Bansback N, Amarsi Z, Laird Birmingham C, Anis AH. The incidence of co-morbidities related to obesity and overweight: a systematic review and meta-analysis. BMC Public Health 2009; 9: 88. doi: 10.1186/1471-2458-9-88
- Reinhardt SL, Boehm R, Blackstone NT, El-Abbadi NH, McNally Brandow JS, Taylor SF, et al. Systematic review of dietary patterns and sustainability in the United States. Adv Nutr 2020; 11: 1016–31. doi: 10.1093/advances/nmaa026
- Lehto E, Kaartinen N, Sääksjärvi K, Männistö S, Jallinoja P. Vegetarians and different types of meat eaters among the Finnish adult population from 2007 to 2017. Br J Nutr 2022; 127: 1060–72. doi: 10.1017/S0007114521001719

*Henna Vepsäläinen

Department of Food and Nutrition PO Box 66,00014 University of Helsinki Helsinki Finland Email: henna.vepsalainen@helsinki.fi